Reviewer #1:

Specific Comments to Authors:

The authors have certainly conducted an interesting and original study with complete statistical studies. This study provides a thorough analysis of postoperative quality of life and pain in patients undergoing surgery for mixed hemorrhoids, offering significant clinical implications. Here are some suggestions:

- The study could benefit from a more information on data analysis. For instance, when using the Student's t-test for inter-group comparisons, it would be beneficial to mention whether analysis of variance (ANOVA) was conducted to validate the reliability of the results.

- In the conclusion section, it would be helpful to emphasize more clearly the clinical practicality of the study and suggest directions for future research.

- Also, it could be clearer in highlighting the generalizability and limitations of the study to aid readers in better understanding the scope of the research.

- Ensure that the cited references are the latest and cover important studies in the relevant field. Overall, the study provides valuable information but has room for further improvement in certain aspects.

Dear Reviewer #1,

Thank you for your valuable feedback and specific comments on our study. We appreciate your recognition of the study's originality and thorough statistical analysis. We have carefully considered your suggestions and will make the following revisions to address them:

Data Analysis: We will provide more information on the data analysis, including whether analysis of variance (ANOVA) was conducted to validate the reliability of the results when using the Student's t-test for inter-group comparisons.

Conclusion Section: We will emphasize the clinical practicality of the study more clearly in the conclusion section and provide suggestions for future research directions.
Generalizability and Limitations: We will make the generalizability and limitations of the study more explicit to help readers better understand the scope of our research.

Updated References: We will ensure that the cited references are the latest and cover important studies in the relevant field.

We appreciate your constructive comments, which will undoubtedly contribute to the improvement of our study. If you have any further suggestions or concerns, please feel free to let us know. We look forward to addressing your feedback and enhancing the quality of our research.

Sincerely,

[Xiao-Wen Sun]